Author's response to reviews

Title: Spatial Distribution and Characteristics of Injecting Drug Users (IDU) in Five Northeastern States of India

Authors:

Gajendra Kumar Medhi (gkmedhi_ro@rediffmail.com)
Jagadish Mahanta (jmahanta@hotmail.com)
Rajatashuvra Adhikary (radhikary@fhiindia.org)
Brogen S Akoijam (brogen@rediffmail.com)
Buno Liegise (bunlieg@yahoo.com)
Kalpana Sarathy Desai (kalpanasarathy13@gmail.com)
Chelliah Joshua Thomas (cjoshuathomas@gmail.com)
Bhupen Sarmah (bsarmah@hotmail.com)

Version: 4 Date: 7 November 2010

Author's response to reviews:

Dear Dr Silvia Martins,

As per your suggestion, we have made our best efforts to improve the style of written English of our article without changing the contents of the manuscript. We have omitted two references from the article (reference No 18 & 27 of earlier article). Hope this time it will be acceptable. We have also formatted the manuscript according to your journal style.

We are ready address if you have any further suggestion on our manuscript.

With best regards

Dr G K Medhi